Author's response to reviews

Title: Predictors of Mortality Among HIV Infected Children on Anti-Retroviral Therapy in Mekelle Hospital, Northern Ethiopia, A retrospective Cohort Study

Authors:

Aregay G G/medhine AG (aregaygebremedhin@gmail.com)
Solomon G G/mariam SG (solomongmariam@yahoo.com)
Fisaha H Haile FH (fisaha.1999@yahoo.com)
Berhe S W/aregawi BW (berheph@yahoo.com)
Carla D Decotelli CD (carladecotelli@gmail.com)

Version: 8 Date: 14 October 2013

Author's response to reviews:

Please find the revised version attached